Author's response to reviews

Title: A randomized controlled trial of sucrose and / or pacifier as analgesia for infants receiving venipuncture in a pediatric emergency department

Authors:

Sarah J Curtis (scurtis@ualberta.ca)
Hsing Jou (hsingjou@yahoo.com)
Samina Ali (draminaali@yahoo.com)
Ben Vandermeer (bv1@ualberta.ca)
Terry P Klassen (TerryKlassen@cha.ab.ca)

Version: 4 Date: 16 July 2007

Author's response to reviews:

Changes to the title page have been made.
Line numbers have been removed.